



Article How Does Agricultural Mechanization Service Affect Agricultural Green Transformation in China?

Xuelan Li^{1,2} and Rui Guan ^{3,*}

- ¹ School of Economics and Management, Anhui Agricultural University, Hefei 230036, China
- ² School of Management, Anhui Science and Technology University, Bengbu 233030, China
- ³ School of Politics and Public Administration, Zhengzhou University, Zhengzhou 450000, China
- * Correspondence: guanrui@zzu.edu.cn

Abstract: Agricultural mechanization service (AMS) is a critical path to achieving agricultural green transformation with smallholders as the mainstay of agricultural production. Based on the panel data of 30 Chinese provinces from 2011 to 2020, this paper measures the AGTFP using the Super-SBM model and examines the effects of different AMS supply agents on AGTFP and spatial spillover effects through the spatial Durbin model. The main conclusions are as follows: First, China's AGTFP showed a stable growth trend, with the mean value increasing from 0.1990 in 2011 to 0.5590 in 2020. Second, the specialization (SPO) and large-scale (LSO) of AMS supply organizations have significantly positive effect on the AGTFP of the local province. However, SPO has a significantly positive effect on the AGTFP of the neighboring provinces, while LSO has the opposite effect. Third, the specialization of AMS supply individuals (SPI) has significantly negative effect on the AGTFP of the local province. In contrast, the large-scale AMS supply individuals (LSI) has the opposite effect. Furthermore, the spatial spillover effects of both are insignificant. Fourth, the spatial spillover effect of AGTFP shows asymmetry among different regions and indicates that AMS resources flow from non-main grain production and economically developed regions to main grain production and less developed regions. These findings provide helpful policy references for constructing and improving the agricultural mechanization service system and realizing the agricultural green transformation in economies as the mainstay of agricultural production.

Keywords: agricultural mechanization; agricultural green transformation; smallholder; spatial spillover effect; China

1. Introduction

Agricultural green transformation is a critical path for the continuous promotion of agricultural modernization and an essential foundation for human society to achieve sustainable development. Since the reform and opening up, China's agricultural modernization process has made remarkable achievements. On the one hand, the productivity of Chinese agriculture has increased significantly [1]. According to statistics, China's total grain production increased from 43.07 million tons in 2003 to 68.24 million tons in 2021, and China has achieved an increase in grain production for eighteen consecutive years. On the other hand, the industrial structure continues to be optimized. The value added of primary industry in China has decreased from 27.7% of GDP in 1978 to 7.3%, gradually approaching the average level of developed countries [2]. However, the rapid development of agricultural modernization in China has been accompanied by excessive use of agricultural chemicals, irrational utilization of agricultural waste, vast consumption of fossil energy, and soil destruction. These problems have led to severe agricultural non-point source pollution and carbon emission [3–6]. Meanwhile, as one of the most populous developing countries in the world, China uses 9% of the global arable land to feed nearly 21% of the global population [7]. The continued expansion of food demand further pushes



Citation: Li, X.; Guan, R. How Does Agricultural Mechanization Service Affect Agricultural Green Transformation in China? *Int. J. Environ. Res. Public Health* **2023**, 20, 1655. https://doi.org/10.3390/ ijerph20021655

Academic Editors: Qiang Gao, Wenjin Long and Jian Chen

Received: 3 December 2022 Revised: 12 January 2023 Accepted: 13 January 2023 Published: 16 January 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). the need for economies to achieve a green revolution centered on sustainable and intensive agricultural development [8–10].

The Chinese government attaches great importance to the agricultural green transformation. In the 13th Five-Year Plan, the word "green" was included in the national development concept for the first time. Subsequently, the Chinese government has repeatedly emphasized the importance of agricultural green transformation in several consecutive Central Government No. 1 documents and programmatic documents such as the 14th Five-Year Plan. It has built specific initiatives for agricultural green transformation to construct high-standard farmland, modern seed industry development, agricultural mechanization promotion, and action to reduce chemical fertilizers. However, the agricultural green transformation in China is complex, and the central problem lies in the pattern of smallholder production and operation formed by the household contract responsibility system. There are about 200 to 300 million smallholder farmers in China, and they have made significant contributions to China's food security as the mainstay of agricultural production [11,12]. However, smallholders are closely associated with land fragmentation, highly dependent on high to excessive factor inputs, and inefficient production [13]. It has to be acknowledged that it is a relatively slow process to solve the problem of agricultural scale operations by land transfer. Currently, the development of the rural land transfer market in China still lags significantly behind the transfer of rural labor, and the proportion and total area of rural land transfer households urgently need improvement [14]. Based on this, socialized agricultural services with agricultural mechanization services (AMS) as the core are considered essential to solving the dilemma of "smallholder management" [15,16].

As early as 500 years ago, "wheat cutters" appeared in China that provided professional wheat-cutting services across hundreds of kilometers, the early agricultural social service in China. Moreover, with the rapid improvement of agricultural mechanization, the AMS system in China has also been rapidly developed. According to statistics, the total output value of the agriculture, forestry, animal husbandry, and fishery service industry snowballed from 287.34 billion yuan in 2011 to 702.98 billion yuan in 2020, of which the income from agricultural machinery operation services in 2020 was as high as 361.503 billion yuan, accounting for 51.42% of the total output value. AMS is essentially a specific form of vertical division of agriculture, which involves smallholders in modern agriculture and replaces land-scale operations with service-scale operations. Furthermore, AMS can break through the paradox of "smallholder management" in which small-scale decentralized operation characteristics cannot endogenize economies of scale [17]. Therefore, exploring the critical role of AMS in the agricultural green transformation can help increase smallholder productivity and reduce adverse environmental impacts to ensure food security. As a result, it can provide important policy implications for the agricultural green transformation in China and other countries with smallholders as the mainstay of agricultural production.

Most studies have visualized agricultural green transformation as agricultural green total factor productivity (AGTPF), and many valuable studies have been conducted in the following directions: (1) Measurement and decomposition of AGTFP. To measure AGTFP, we need to construct an indicator system from two dimensions: input and output. Input indicators mainly include labor, land, machinery, fertilizer, and agricultural energy consumption [18-20]. In the output dimension, compared with the traditional total factor productivity in agriculture, the measurement of AGTFP incorporates undesired outputs into the indicator system, mainly including carbon emissions and agricultural non-point source pollution [21]. However, some scholars argue that it is questionable whether livestock farming and organic solid waste should be considered as pollution in agricultural non-point source pollution [22]. On this basis, the decomposition of AGTFP reveals that technological progress is considered the driving force for the continuous growth of AGTFP in China [23], while declining technical efficiency and scale efficiency slow down the growth of AGTFP [24]. (2) Spatiotemporal characteristics of AGTFP. Existing studies have reached a certain consensus on the spatiotemporal evolution characteristics of AGTFP in China. Since the reform and opening up, China's AGTFP has shown a fluctuating growth trend, but

the gaps among regions have also gradually expanded [25–27]. AGTFP in eastern, central, and western China decreases in order [22,28,29]. (3) Analysis of the driving mechanism of AGTFP. Changes in factor inputs directly contribute to AGTFP, including farm size expansion, green manure application, agricultural mechanization increase, and human capital input [30–33]. At the same time, policy interventions are also essential to address factor input overload and environmental damage externalities. Existing studies have empirically tested that implementing crop insurance, carbon trading systems, and other agricultural environment protection policies can help increase AGTFP [18,19,34]. In addition, the use of new technologies is also a significant driving force for the sustained growth of AGTFP, including renewable energy consumption, the spread of Internet technologies, and the development of inclusive digital finance [35–37]. Some scholars have also studied the influencing factors that hinder the growth of AGTFP, including factor market distortion, resource mismatch, farmland pollution, agricultural disasters, and climate change [38–41].

In addition, some studies have also explored the critical role of agricultural services in promoting agricultural green transformation. Xu et al. characterized agricultural services in terms of agricultural services output and indicated that agricultural services helped to promote the growth of AGTFP, and the improvement effect was higher in the eastern region [42]. Zhu et al. characterized productive agricultural services in terms of the number of people employed in agricultural services and indicated that productive agricultural services effectively improved agri-environmental efficiency by optimizing inputs and increasing outputs [43]. From a micro perspective, Li et al. showed that adopting green agricultural services by farmers helped improve AGTFP, and the improvement effect was more substantial with higher adoption [44]. Zhang et al. showed that fertilizer application services and pest control services were crucial aspects of promoting AGTFP [45]. From a comprehensive perspective, studies have recognized the critical role of agricultural services in the agricultural green transformation. However, few studies have directly focused on the impact of AMS on the agricultural green transformation. At the same time, compared with other agricultural services, the "cross-regional operation" feature of AMS has also been neglected, which will underestimate the spatial spillover effect of AMS in promoting agricultural green transformation. In addition, AMS has multiple supply agents. However, the existing studies have not considered this reality, so there is a lack of in-depth research on the development direction of AMS with the goal of agricultural green transformation.

Given the limitations of existing studies, the research questions of this paper are as follows. (1) Whether there is an impact and spatial spillover effect of AMS on AGTFP. (2) Whether the development direction of different AMS agents is the same under the goal of agricultural green transformation. (3) Is there heterogeneity in the spatial spillover effect of AMS among different regions? To solve these problems, this paper measures the AGTFP using the Super-SBM model based on the panel data of 30 provinces from 2011 to 2020 in China. Based on this, the spatial Durbin model tests the influence of the degree of specialization and large-scale of different AMS supply agents on the AGTFP and the spatial spillover effect. Furthermore, this paper uses the two-regime spatial Durbin model to test the asymmetry of the spatial spillover effect between different regimes. This paper aims to provide theoretical support for the agricultural green transformation in economies with smallholders as the mainstay of agricultural production.

The rest of this paper is organized as follows. Section 2 provides a literature review and constructs a theoretical framework. Section 3 presents the methodology, empirical models, and data sources. Section 4 describes the empirical results and findings. Section 5 provides further discussion and limitations. Section 6 provides the main conclusion.

2. Literature Reviews and Theoretical Framework

2.1. Literature Reviews

Adam Smith proposed early in the Wealth of Nations that the division of labor could lead to gains from specialization and emphasized that market size determines the division of labor. However, because of the lack of scale, the division of labor in agricultural production tends to be less prevalent than in industrial production. Under this view, Otsuka proposed that only in large farms can investments in mechanization generate adequate returns for operators. Therefore, he argued that the specialized division of labor in Chinese agricultural production would be challenging due to the scale of agriculture and that labor productivity growth would be prolonged [46]. However, the practice of agricultural development in China shows that with an average agricultural size of about 0.5 ha, China has developed a division of labor and experienced rapid agricultural mechanization. For this reason, the rapid development of AMS has facilitated the organic combination of the provision of specialized labor and the services of large harvesting machines, thus effectively increasing agricultural productivity by achieving economies of scale in services [47].

Adequate research has been conducted regarding the factors influencing AMS purchase decisions and the economic and social effects they produce. Specifically, the operation scale is an essential factor influencing the adoption of AMS by farmers. Due to the vast sunk costs and high maintenance costs of agricultural machinery, it is often not cost-effective for smallholders who lack the capital to purchase machinery on their own [17,48]. Therefore, small farmers tend to prefer to purchase AMS. In contrast, large-scale operators tend to purchase machinery on their own and have a tendency to further transform into agricultural mechanization service providers [49]. In addition to the scale of operation, non-farm employment income, education level, and farming experience are influential factors affecting farmers' purchase of AMS [50–52]. Moreover, at the macro level, the level of economic development, population size and agricultural machinery purchase subsidy policy have an important impact on the level of AMS in each province [53]. Some scholars have explored the impact of purchasing AMS on farmers' agricultural production. Deng et al. found that the productivity of farmers who purchased AMS increased by 25.61% in a sample of farmers in Shandong [54]. A study by Tang et al. showed that agricultural services helped farmers reduce their production costs and technical services had the most significant impact, followed by processing services and AMS [55]. The study by Qiu et al. further indicated that the increase in productivity after purchasing AMS was significantly higher in medium-sized farms than in small and large farms [56]. Some scholars have also explored the impact of AMS adoption on farm household welfare. Among them, Mi et al. used a sample of cotton farmers in Xinjiang and found that the adoption of AMS by small farmers significantly increased household income, consumption expenditure, and off-farm employment opportunities [57]. Lyne et al. found that the extension of AMS in South Africa contributed to higher household agricultural income and helped farmers improve the quality of their agricultural products [58]. However, the breakdown and poor maintenance of agricultural machinery will reduce the profitability of AMS supply agents and further weaken the improvement of AMS to the welfare of farm households [59]. In addition, excessive AMS prices will also lead to the withdrawal of smallholder farmers from agricultural production, which in turn will negatively impact smallholder welfare [60]. In general, existing studies have been valuable and well-explored around AMS. However, a suitable theoretical framework for the mechanism of action between AMS and AGTFP has yet to be established.

2.2. Theoretical Framework

By combing the existing studies, this paper constructs a theoretical framework of the impact of AMS on AGTFP from four aspects, including factor allocation, planting structure, technological progress, and spatial spillover effect, as shown in Figure 1. AMS affects AGTFP by changing the allocation of agricultural production factors, mainly resource inputs (labor, land, agricultural machinery) and environmental inputs (pesticides, fertilizers, agricultural films). From the perspective of resource factor inputs, the rapid development of AMS services will directly increase the input scale of agricultural machinery factors, which will directly cause the expansion of agricultural energy consumption and bring potential problems for the green transformation of agricultural labor through the labor substitution

effect, which expands the land operation scale [48,53]. Furthermore, economies of scale help to achieve increased agricultural productivity and reduce agricultural carbon emissions. In terms of environmental factor inputs, AMS can improve farmers' expectations of crop yields, which in turn effectively reduces excessive inputs of environmental factors such as pesticides and fertilizers through factor substitution mechanisms [45,61]. Additionally, AMS can help promote the formation of pro-environmental behaviors among farmers, which in turn improves the AGTFP at the micro level [44]. However, it has also been pointed out that AMS enhances the use of abandoned or poorly managed arable land, thereby increasing the intensity of environmental factor inputs such as pesticides [62].



Figure 1. The theoretical Framework for the impact of AMS on AGTFP.

AMS affects AGTFP by changing the cropping structure. Cash crops are more laborintensive than food crops, and AMS constrains the "non-grain trend " of crops by facilitating the transfer of agricultural labor. At the same time, food crops are more suitable for mechanization than cash crops, so with the increase in labor cost and the decrease in AMS price, farmers will prefer to grow food crops. In addition, the development of AMS will increase the comparative advantage of food crops by improving the expansion of the agricultural operation scale, thus promoting a "food-based" cropping structure [11]. Compared with cash crops, food crops have a smaller scale of production factor inputs and a more significant carbon sink effect. Therefore, substituting food crops for cash crops will help increase the AGTFP of agricultural systems.

AMS affects AGTFP through technological advances. Firstly, applying AMS can effectively improve the efficiency of all types of energy use in agricultural production, thereby reducing agricultural energy inputs in the input dimension and undesired outputs such as agricultural carbon emissions. Secondly, the rapid development of AMS can also replace traditional agricultural energy with new and renewable energy sources, thus optimizing energy use structure [63]. Thirdly, AMS can bring advanced organizational management experience to agricultural producers and deepen the specialization of agricultural producers through vertical division of labor, which contributes to the improvement of agricultural production organization, management level, and production efficiency [64]. Fourthly, AMS can help increase the timeliness of operations in the agricultural production chain, avoiding the impact of work-hour delays on crop growth, thus indirectly increasing output.

The spatial spillover effect of AMS. Compared with other agricultural services, the cross-regional operation is an important feature of AMS. The vast territory of China and the apparent difference in crop production cycles between regions provide the possibility for the cross-regional operation of AMS. At the same time, the cross-regional operation also helps to expand the market scale and further deepen the degree of vertical division of labor in the whole agricultural system, which is more conducive to the realization of service economies of scale [65]. The cross-regional operations of AMS are mainly driven by large- and medium-sized agricultural machinery. They operate between regions within a

day's economic distance, and operations' intensity increases yearly [66]. Meanwhile, due to the high seasonal requirements of crop production, AMS agents often operate across regions between different latitudes [67]. Therefore, through cross-regional operations, AMS can influence the allocation of production factors, cropping structure, and technological progress in neighboring regions, changing the AGTFP of the whole agricultural system.

3. Methodology and Data

3.1. Measurement of AGTFP

AGTFP is an important indicator of the agricultural green transformation. Currently, productivity is widely measured using the non-parametric approach. In a non-parametric method such as envelopment analysis (DEA), the efficiency of a decision-making unit is described by the relationship between inputs and outputs on a linear piecewise frontier constructed by the DEA model [23,68–70]. In addition, some scholars have combined DEA models with machine learning to optimize the productivity measures [71,72]. In the early research, the method requires the selection of input and output angles of the model and the same proportional variation of inputs or outputs, thus making it difficult to match its measurement results with the actual situation. Therefore, considering the non-angle and non-radial characteristics of the DEA model, the super-SBM model is proposed. Based on this, this paper incorporates agricultural carbon emissions as non-expected outputs into the super-SBM model to measure the AGTFP. It is assumed that the decision unit k has input vectors $x \in R^M$, desired output vectors $y^g \in R^{s_1}$, and undesired output vectors $y^b \in R^{s_2}$, respectively. For the decision unit k to be measured, as in Equation (1):

$$\rho = \min \frac{1 + \frac{1}{m} \sum_{i=1}^{m} \frac{s_{i}}{x_{ik}}}{1 - \frac{1}{s_{1} + s_{2}} \left(\sum_{r=1}^{s_{1}} \frac{s_{r}^{s_{1}}}{y_{rk}^{s_{r}}} + \sum_{t=1}^{s_{2}} \frac{s_{t}^{b}}{y_{rk}^{b}} \right)} \\$$
s.t. $\sum_{j=1, j \neq k}^{n} x_{ij}\lambda_{j} - s_{i}^{-} \leq x_{ik}$

$$\sum_{j=1, j \neq k}^{n} y_{rj}\lambda_{j} + s_{r}^{g} \leq y_{rk}^{g}$$

$$\sum_{j=1, j \neq k}^{n} y_{tj}\lambda_{j} - s_{t}^{b} \leq y_{tk}^{b}$$

$$\lambda \geq 0, \ s^{g} \geq 0, \ s^{b} \geq 0, \ s^{-} \geq 0$$
(1)

where ρ represents the AGTFP under the super-SBM model, which can be greater than 1, so that the effective decision unit can be distinguished, λ is the weight vector, and s^{g} , s^{b} , and s^{-} are slack variables.

From the model set, it is clear that the input and output variables selection is critical for the measurement of AGTFP. Concerning existing studies, this paper constructs an indicator system for measuring AGTFP, as shown in Table 1. In contrast to existing studies, this paper replaces the irrigated area factor with the amount of agricultural water consumption. This is because both irrigated area and land input elements are characterized using land area, and it is unreasonable to put both in the same model. Meanwhile, due to the significant differences in farming practices and crop cultivation structures in different regions, the weight of agricultural water consumption per unit of irrigated area is different. Therefore, it is impossible to truly reflect the amount of water resources input through the irrigated area [22]. In addition, this paper selects agricultural carbon emissions as undesired output. Referring to existing studies [73–76], this paper defines the sources of agricultural carbon emissions as pesticides, agricultural fertilizers, agricultural plastic films, agricultural diesel fuel, agricultural machinery, and irrigation. The carbon emission factors of each type of carbon source refer to IPCC [77].

Indicator	Index	Definition	Mean	Unit
	Labor input	Number of employees in agriculture	780.16	10,000 peoples
	Land input	The sowing area of crops	5336.93	1000 HA
Mech ing	Mechanical input	Total power of agricultural machinery	3318.60	10,000 kW
Input	Fertilizer input	Chemical fertilizer application in agriculture	185.89	10,000 tons
indicators	Film input	Amount of Agricultural plastic film application	7.98	10,000 tons
	Pesticide input	Amount of pesticide application	5.35	10,000 tons
	Energy input	Amount of Agricultural diesel consumption	66.75	10,000 tons
	Water input	Amount of Agricultural water consumption	121.89	100 million m ³
Output	Desired	The gross production of agriculture	1784.42	100 million yuan
indicators	Undesired output	Agricultural carbon emissions	29,416.66	10,000 tons

Table 1. Indicator system construction for AGTFP.

3.2. Variables Description

3.2.1. Core Explanatory Variables

Based on the two research scales of organization and individual, this study deconstructs the development of AMS into large-scale and specialization, from which four core explanatory variables are constructed. At the organizational level, it mainly includes largescale AMS supply organizations (LSO) and specialized AMS supply organizations (SPO). The former is indicated by the ratio of the number of AMS organizations with agricultural machinery with the original value of 500,000 yuan or more to the number of total AMS organizations; the ratio of the number of agricultural machinery specialized cooperatives to the number of total AMS organizations indicates the latter. From the individual level, it mainly includes large-scale AMS supply individuals (LSI) and specialized AMS supply individuals (SPI). The former is indicated by the ratio of the number of AMS farmers with agricultural machinery with the original value of 200,000 yuan or more to the number of total AMS farmers; the ratio of AMS specialized farmers to the number of total AMS farmers; the ratio of AMS specialized farmers to the number of total AMS farmers indicates the latter.

As shown in Figure 2, the development of AMS in China shows the primary trend of large-scale specialization. From the organization level, while the number of AMS organizations steadily increased from 2011 to 2020, LSO continued to grow from 11.51% to 30.40%, and SPO continued to grow from 16.33% to 38.72%. However, at the individual level, while the number of AMS farmers declined, LSI slightly increased from 1.08% to 1.77%, and SPI decreased from 12.45% to 10.53% in 2011–2020. The weakening of AMS at the individual level is due to the massive shift of rural labor, with some AMS farmers moving to non-farm sector employment. It may also be due to the structural shift of some AMS farmers from individual to organizational through cooperatives and acquisitions.



Figure 2. Development of agricultural mechanization services.

3.2.2. Control Variables

This paper selects the following control variables concerning the existing literature: (1) Degree of industrialization (IND), the ratio of value added of secondary industry to total output value; (2) Per capita disposable income of rural residents (lnDI), in yuan; (3) Agricultural fiscal expenditure (lnAFE), in billion yuan; (4) Area affected, in thousand hectares; (5) Urbanization rate (UR), the ratio of the number of permanent urban residents to the number of permanent residents at the end of the year in the province; (6) GDP per capita (lnGDP), the ratio of gross regional product to the number of permanent residents at the end of the year; (7) Mechanization intensity (MI), the ratio of total agricultural machinery power to sown area; (8) Population (POP), the number of people registered at the end of the year. To exclude the effect of prices, the relevant data are converted to constant prices with 2011 as the base period. Descriptive statistics are shown in Table 2.

Table 2. Descriptive statistics of control variables.

Variable	Mean	Std. Dev.	Max	Min
IND	0.337	0.080	0.100	0.574
lnDI	9.354	0.418	8.271	10.461
InAFE	6.160	0.573	4.519	7.200
InDIS	5.985	1.550	0.693	8.349
UR	59.006	12.218	35.030	89.600
lnGDP	10.841	0.436	9.706	12.013
MI	0.328	0.112	0.136	0.693
POP	4599.783	2837.845	568.000	12,624.000

3.3. Empirical Models

3.3.1. Spatial Durbin Model

In order to study the effects of different AMS supply agents on AGTFP and spatial spillover effects, the following spatial Durbin model (SDM) was constructed for empirical analysis.

$$y_{it} = \rho \sum_{i=1}^{n} W_{ij} y_{it} + \beta_1 AMS_{it} + \delta_1 \sum_{i=1}^{n} W_{ij} AMS_{it} + \beta_2 X_{it} + \delta_2 \sum_{i=1}^{n} W_{ij} X_{it} + \mu_i + \gamma_t + \varepsilon_{it}$$
(2)

In Equation (2), y_{it} is the explanatory variable, which indicates the observed value of AGTFP of the province *i* in year *t*; *AMS*_{*it*} is the core explanatory variable, including LSO and SPO at the organizational level and LSI and SPI at the individual level; X_{it} is the control variable, including the eight control variables mentioned above; W_{ij} indicates the spatial weight matrix, and in order to ensure the robustness of the research results, the spatial adjacency matrix, geographic distance matrix, and economic distance matrix are constructed for spatial econometric analysis, respectively; ρ is the spatial autocorrelation coefficient, which indicates the impact of AGTFP of neighboring provinces on the local region; β_1 and β_2 are the coefficient to be estimated for the core explanatory variables and control variables, respectively; δ_1 and δ_2 are the estimation coefficient of the spatial lag term of the core explanatory variables and control variables, respectively; μ_i and γ_t indicate the individual effect and time effect, respectively, and ε_{it} is the random disturbance term.

3.3.2. Two-Regime Spatial Durbin Model

The cross-regional operations of AMS organizations and individuals are driven by demand and revenue and thus exhibit asymmetrical characteristics. On the one hand, the Chinese government set 13 provinces, including Anhui, Shandong, Henan, etc., as the main grain production areas in 2001 to ensure food security. The main grain-production areas undertake important grain production tasks, so their demand for AMS is strong. Meanwhile, developed regions such as Beijing, Shanghai, and Tianjin have higher prices for AMS, which are also more attractive to the flow of AMS organizations and individuals. Therefore, referring to the study of Elhorst [78], this paper constructs a two-regime Spatial Durbin Model to test the heterogeneity effect of AMS to AGTFP. The specific settings of the model are as follows:

$$y_{it} = \rho_{d=1}d_{it}\sum_{i=1}^{n}W_{ij}y_{it} + \rho_{d=0}(1-d_{it})\sum_{i=1}^{n}W_{ij}y_{it} + \beta_1AMS_{it} + \delta_1\sum_{i=1}^{n}W_{ij}AMS_{it} + \beta_2X_{it} + \delta_2\sum_{i=1}^{n}W_{ij}X_{it} + \mu_i + \gamma_t + \varepsilon_{it}$$
(3)

In Equation (3), d_{it} is a binary dummy variable that distinguishes different regime, and the specific replication rules are as follows. The other coefficients remain consistent with Equation (2).

$$d_{it}^{1} = \begin{cases} 1, & \text{main grain producing area} \\ 0, & \text{else} \end{cases}$$
(4)

$$d_{it}^2 = \begin{cases} 1, & \text{developed area} \\ 0, & \text{else} \end{cases}$$
(5)

3.4. Data Resources

Since the SBM model is sensitive to abnormal data and there are differences in the statistical caliber of some regions, the data of Tibet, Hong Kong, Macau, and Taiwan provinces are excluded. Finally, this study constructs the panel data of 30 provinces in China from 2011 to 2020. Specifically, the core explanatory variable AMS is obtained from the *China Agricultural Machinery Industry Yearbook* (2012–2021); the data on agricultural value added, total agricultural machinery power, crop sown area, agricultural fertilizer application, agricultural plastic film use, agricultural diesel fuel, and pesticide use used in the calculation of AGTFP are taken from the *China Rural Statistical Yearbook* (2012–2021); the number of employees in the primary industry is obtained from the WIND database; other control variables such as regional GDP and agricultural-related fiscal expenditure are taken from the *China Statistical Yearbook* (2012–2021).

4. Results

4.1. Temporal Evolutionary Characteristics of AGTFP in China

Table 3 shows the AGTFP of China's provinces from 2011 to 2020. Overall, China's AGTFP shows a stable upward trend, with the average value increasing from 0.1990 in 2011 to 0.5590 in 2020 (line 2), with an average annual growth rate of 13.42%. Between 2011 and 2020, Jilin (line 13) and Inner Mongolia (line 11) have the lowest AGTFP averages of 0.1449 and 0.1549, respectively, and Jilin is also one of the provinces with the slowest AGTFP growth rate of 4.08% per year. Beijing (line 7) and Hainan (line 27) have the highest average value of AGTFP, with 0.6385 and 0.5887, respectively. At the same time, Tianjin (line 8) and Ningxia (line 35) have the fastest growth rate, with an average annual growth rate of 24.28% and 27.22%, respectively.

Table 3. The measurement results of AGTFP in China, 2011–2020.

Province	2011	2012	2013	2014	2015	2016	2017	2018	2019	2020
Whole	0.1990	0.2168	0.2353	0.2489	0.2620	0.2874	0.3094	0.3534	0.4327	0.5590
MGP	0.1799	0.1938	0.2083	0.2157	0.2229	0.2337	0.2416	0.2572	0.2827	0.3567
NMGP	0.2136	0.2344	0.2560	0.2743	0.2919	0.3285	0.3612	0.4269	0.5474	0.7138
EDR	0.2510	0.2733	0.2989	0.3154	0.3346	0.3684	0.4026	0.4595	0.5887	0.6812
UEDR	0.1730	0.1885	0.2035	0.2157	0.2257	0.2469	0.2628	0.3003	0.3547	0.4979
Beijing	0.3095	0.3768	0.4432	0.4943	0.5720	0.6139	0.6896	0.8063	1.0572	1.0217
Tianjin	0.1607	0.1694	0.1829	0.1935	0.2013	0.2087	0.2319	0.3130	0.3422	0.4642
Hebei	0.1655	0.1779	0.1935	0.1894	0.1859	0.2014	0.2142	0.2430	0.2602	0.3007
Shanxi	0.1366	0.1451	0.1559	0.1628	0.1564	0.1738	0.1937	0.2029	0.2180	0.2582
Inner Mongolia	0.1388	0.1457	0.1567	0.1558	0.1496	0.1544	0.1460	0.1561	0.1660	0.1803
Liaoning	0.2074	0.2290	0.2400	0.2431	0.2689	0.2772	0.2851	0.3235	0.3659	0.3925
Jilin	0.1445	0.1521	0.1529	0.1525	0.1470	0.1305	0.1199	0.1332	0.1429	0.1737
Heilongjiang	0.1418	0.1726	0.2123	0.2251	0.2234	0.2261	0.2399	0.2536	0.2823	0.3050
Shanghai	0.3783	0.3894	0.4177	0.4117	0.4089	0.3600	0.3548	0.4444	0.4600	0.4356
Jiangsu	0.2197	0.2442	0.2566	0.2678	0.2994	0.3032	0.3100	0.3119	0.3283	0.3973
Zhejiang	0.2261	0.2447	0.2791	0.3005	0.3139	0.3598	0.4334	0.4496	0.5377	0.5467
Anhui	0.1330	0.1431	0.1506	0.1615	0.1623	0.1690	0.1820	0.1840	0.1978	0.2169
Fujian	0.2892	0.3233	0.3445	0.3760	0.3976	0.5018	0.5953	0.7180	1.0882	1.0268
Jiangxi	0.1290	0.1380	0.1822	0.1910	0.2117	0.2303	0.2339	0.2466	0.2695	0.2887
Shandong	0.1936	0.1999	0.2262	0.2402	0.2495	0.2531	0.2591	0.2767	0.2929	0.3163
Henan	0.1897	0.2002	0.2066	0.2297	0.2290	0.2353	0.2434	0.2652	0.2944	0.3475
Hubei	0.2231	0.2379	0.2488	0.2527	0.2573	0.2907	0.3042	0.3149	0.3530	0.3912
Hunan	0.2003	0.2080	0.2030	0.2059	0.2061	0.2203	0.2322	0.2401	0.2880	0.3265
Guangdong	0.2918	0.3120	0.3413	0.3557	0.3733	0.4705	0.5106	0.5865	0.7635	1.1907
Guangxi	0.2183	0.2233	0.2348	0.2442	0.2541	0.2795	0.3054	0.3301	0.4389	0.4881
Hainan	0.3048	0.3362	0.3490	0.3946	0.4270	0.5433	0.5923	0.8344	1.0507	1.0547
Chongqing	0.2178	0.2355	0.2488	0.2616	0.2725	0.3220	0.3372	0.3732	0.6644	1.0216
Sichuan	0.2528	0.2702	0.2785	0.2898	0.3076	0.3461	0.3712	0.3951	0.4344	1.0000
Guizhou	0.1585	0.1984	0.2248	0.2934	0.3967	0.4520	0.5047	0.5756	0.7185	1.0977
Yunnan	0.1424	0.1600	0.1794	0.1899	0.1915	0.2016	0.2115	0.2824	0.4057	1.0000
Shaanxi	0.2495	0.2671	0.2943	0.3203	0.3210	0.3548	0.3812	0.4548	0.5889	1.0372
Gansu	0.1008	0.1087	0.1158	0.1184	0.1229	0.1403	0.1629	0.1803	0.2065	0.3193
Qinghai	0.1539	0.1743	0.2014	0.2057	0.1989	0.2219	0.2342	0.2579	0.3168	0.4640
Ningxia	0.1197	0.1285	0.1402	0.1468	0.1633	0.1787	0.1902	0.2167	0.2102	0.3456
Xinjiang	0.1727	0.1924	0.1984	0.1935	0.1912	0.2021	0.2116	0.2308	0.2387	0.3620

Note: MGP is the main grain production area; NMGP is the non-main grain production area; EDR is the economically developed region; NEDR is the economically developing region.

By 2011–2015 China's 12th Five-Year Plan period (column 2 and column 6, line 2), the national AGTFP average value increased from 0.1990 to 0.2620 with an average annual growth rate of 7.51%. 2016–2020 is China's 13th Five-Year Plan period (column 7 and column 11, line 2). The national average value of AGTFP increases from 0.2874 to 0.5590, with an average annual growth rate of 18.15%. In comparison to the "12th Five-Year" period, China's AGTFP growth rate and magnitude are higher. The reasons for this are, first, the

development concept differences. During the "12th Five-Year Plan" period, the main goal of China's agricultural production was to ensure the adequate supply of food and other major agricultural products. In the "13th Five-Year Plan" period, the basic principle of China's agricultural production is to adhere to sustainable development, and the "13th Five-Year Plan" clearly proposes to "take the path of output efficiency, resource conservation, and environmental friendliness." Second, the difference in development basis. During the "12th Five-Year Plan" period, China still faced the difficulties of weak agricultural infrastructure, low material equipment, and lagging social services. At the beginning of the "13th Five-Year Plan", China built 400 million mu of high-standard farmland, and the contribution rate of agricultural science and technology progress and the total mechanization rate of crop cultivation, planting, and harvesting have reached 56% and 63%, respectively.

From the perspective of different regions, the degree and speed of agricultural green transformation in the main grain production areas are slightly lower than the non-main grain production areas (line 3 and line 4), with the average AGTFP values of 0.1799 and 0.3567 in 2011 to 0.2136 and 0.7138 in 2020, respectively, and the average annual growth rates of 11.61% and 14.81%, respectively. This is because the main grain production areas are critical to ensuring national food security, and increasing total grain production areas will be the main battlefield for the agricultural green transformation in China. The mean value of AGTFP in economically developed regions (line 5), represented by large urban agglomerations such as Beijing-Tianjin-Hebei, Yangtze River Delta, and Pearl River Delta, is 0.2510 in 2011 and 0.6812 in 2020, significantly higher than 0.1730 in 2011 and 0.4979 in 2020 in economically developing regions (line 6). Economically developed regions have a higher level of agricultural modernization and a noticeable technology diffusion effect, thus promoting the rapid growth of AGTFP.

4.2. Applicability Test of the Spatial Econometric Model

Before using the spatial model regression, the spatial autocorrelation test needs to be performed on AGTFP. As seen from Figure 3, the Moran's I of AGTFP is between 0.138 and 0.229, and the Moran's I of each year is significant at the 5% statistical level. It indicates a significant spatial autocorrelation of AGTFP in China, which is suitable for regression analysis using the spatial econometric model. Specifically, during the 12th Five-Year Plan period (2011–2015), the AGTFP of each province in China showed a decreasing trend and fell to the lowest point of 0.138 in 2015. It can be concluded that the inter-provincial AGTFP in China has clustering characteristics, i.e., high values are clustered with high values, and low values are clustered with low values. However, the clustering characteristics vary widely between years and are relatively unstable overall. This also indicates that there are still significant differences in AGTFP between provinces, and there is more room for development.

On this basis, this paper adopts a series of tests to determine the applicability of the spatial Durbin model. First, in the LM test, the LM statistics of the spatial lag model and the spatial error model are 146.754 and 229.790, respectively, and both are significant at the 1% statistical level. This indicates that the original hypothesis of no spatial lag and the spatial error should be rejected, and the spatial panel model should be used for the empirical analysis. Second, in the LR test, the statistical value is 47.860 when the spatial Durbin model is compared with the spatial error model and 31.360 when the spatial Durbin model is compared with the spatial Durbin model cannot be reduced to a spatial error model and a spatial lag model in this study. Finally, the Hausman test has a statistical value of 29.120, which is significant at the 1% statistical level. This indicates that the original hypothesis of using random effects should be rejected, and a fixed effects spatial Durbin model should be used.



Figure 3. Moran's I of Agricultural Green Total Factors Productivity in China, 2011 to 2020.

4.3. The Impact of AMS Supply Organizations on AGTFP

Table 4 demonstrates the effect of AMS supply organization on AGTFP. Under the three spatial weight matrices, the spatial autoregressive coefficient of AGTFP is positive and significant at the 1% statistical level for both LSO and SPO as the core explanatory variables, which is consistent with the studies of Xiao et al. [23] and Ma et al. [79]. This suggests that the AGTFP of neighboring provinces or provinces with similar degrees of economic development can effectively contribute to the increase in AGTFP of the local province. The possible influence mechanism is the inter-regional technology spillover effect and demonstration effect. At the same time, this finding is consistent with the previous spatial autocorrelation test and further reflects the validity of the findings of this study.

Regarding the main effects, the regression coefficients of LSOs ranged from 0.132–0.172, all of which were statistically significant at the 10% level. The regression coefficients of SPO ranged from 0.194–0.255, all of which were significant at the 1% statistical level. These results show that increasing the scale and specialization of AMS organizations in the local region can effectively increase AGTFP and thus promote the green transformation of agriculture. Specifically, the expansion of the AMS organization scale helps to increase agricultural productivity and effectively restrain excessive input of environmental factors through the factor substitution mechanism, thus increasing AGTFP. At the same time, the increase in AMS organization specialization helps to accelerate the technological progress in agricultural production, promoting agricultural green transformation.

Regarding the control variables, the expansion of the affected area will lead to a decrease in food production and agricultural output, resulting in a significant decrease in AGTFP. As for the micro-level explanation, after suffering from natural disasters, in order to reduce losses to avoid falling into poverty, farmers and other operators will increase factor inputs such as fertilizers and pesticides on non-affected plots, which increases undesired output. Meanwhile, higher regional GDP means better regional economic development and agricultural infrastructure, promoting a significant increase in AGTFP. This finding also provides supporting evidence for the later paper to explore the effect of AMS on AGTFP from the level of economic development. In addition to this, the intensity of agricultural mechanization and population size will also significantly increase AGTFP, which is consistent with Zhu et al. [31] and Chi et al. [80].

_

	И	/1	V	V2	И	/3
ISO	0.164 *		0.132 *		0.172 **	
L30	(0.090)		(0.071)		(0.083)	
SPO		0.228 ***		0.255 ***		0.194 ***
510		(0.057)		(0.077)		(0.064)
IND	-0.703 **	-0.099	-0.432	-0.018	-0.508 *	-0.006
1,2	(0.322)	(0.303)	(0.311)	(0.292)	(0.308)	(0.286)
lnDI	-0.070	-0.309	-0.192	-0.351 *	-0.240	-0.429 **
	(0.208)	(0.204)	(0.200)	(0.194)	(0.203)	(0.195)
lnAFE	-0.076	-0.077	-0.054	-0.032	-0.093 *	-0.055
	(0.049)	(0.050)	(0.048)	(0.047)	(0.050)	(0.050)
Indisaster	-0.015 *	-0.016 *	-0.015 *	-0.016 **	-0.018 **	-0.232 ***
	(0.009)	(0.009)	(0.008)	(0.008)	(0.008)	(0.008)
UR	0.000	0.006	-0.001	-0.001	0.000	0.002
	(0.005)	(0.005)	(0.005)	(0.005)	(0.005)	(0.005)
lnGDP	0.303 ***	0.261 ***	0.288 ***	0.247 ***	0.255 ***	0.230 ***
	(0.075)	(0.073)	(0.071)	(0.064)	(0.069)	(0.064)
MI	0.321^{444}	0.297 ***	0.320 ***	0.300 ***	(0.377^{333})	0.337 ***
	(0.117)	(0.112)	(0.110)	(0.106)	(0.113)	(0.110)
POP	0.000^{44}	0.000 *	0.000	0.000 **	0.000 ***	0.000
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
LSO*W	-0.437		-1.002		-1.119	
	(0.107)	0 370 ***	(0.073)	1 765 ***	(0.507)	0 877 **
SPO*W		(0.37)		(0.639)		(0.402)
	0.433	(0.140)	_3 737	(0.039) -1.859	2 164	(0.402)
IND*W	(0.43)	(0.637)	(2.823)	(2730)	(2.104)	(2.106)
	0.086	0 279	0 701	1 044	0.095	0.605
lnDI*W	(0.382)	(0.369)	(1, 229)	(1 135)	(0.887)	(0.791)
	0.315 ***	0.311 ***	1 479 ***	1 808 ***	0.287	0 470 **
lnAFE*W	(0.098)	(0.099)	(0.357)	(0.344)	(0.231)	(0.201)
	-0.006	0.010	0.028	0.734	-0.022	0.030
Indisaster*W	(0.017)	(0.016)	(0.054)	(0.055)	(0.034)	(0.032)
TIDUTAT	-0.045 ***	-0.047 ***	-0.027	-0.022	-0.006 *	-0.016
UR*W	(0.011)	(0.011)	(0.029)	(0.028)	(0.021)	(0.019)
	0.224 *	0.357 ***	-0.196	0.571	-0.062	0.510
InGDP*W	(0.134)	(0.110)	(0.529)	(0.407)	(0.389)	(0.333)
N #T#TA7	-0.091	0.135	0.471	0.767	-0.852	0.468
IVII ^a VV	(0.232)	(0.225)	(0.695)	(0.680)	(0.689)	(0.630)
	0.000 *	0.000	0.000	0.000	0.000	0.000
POP [*] W	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
rh o	0.232 **	0.289 ***	1.638 ***	1.845 ***	0.728 ***	0.802 ***
1110	(0.093)	(0.090)	(0.294)	(0.283)	(0.209)	(0.204)
Province FE	YES	YES	YES	YES	YES	YES
Time FE	YES	YES	YES	YES	YES	YES
R-squared	0.261	0.235	0.444	0.511	0.245	0.373
Obs	300	300	300	300	300	300

Table 4. The impact of AMS on AGTFP in China: the organizational level.

Note: Standard errors in parentheses, *, ** and *** indicate significant at the 10%, 5% and 1% statistical levels, respectively.

In terms of the spatial lagged variables, the regression coefficients of LSO*W ranged from -0.437 to -1.662 and were all significant at the 5% statistical level, i.e., there was a significant negative spatial spillover effect of the effect of LSO on AGTFP. On the other hand, the regression coefficients of SPO*W ranged from 0.379 to 1.265, and all were significant at the 1% statistical level, i.e., there was a significant positive spatial spillover effect of the effect of SPO on AGTFP. The spatial spillover effect is significant precisely because both LSOs and SPOs are the supplying agents of agricultural machinery cross-regional operation services. In addition, the regression coefficient of lnAFE*W is significantly positive, indicating that the expansion of the scale of agriculture-related expenditures in neighboring provinces will contribute to the increase in AGTFP in this province, a similar finding to that of Xiao et al. [23].

On this basis, the spatial effects of the core explanatory variables are further decomposed, and the results are shown in Table 5. Taking the spatial adjacency weight matrix (W1) as an example, the direct effects of LSO and SPO on AGTFP are significantly positive at the 5% and 1% statistical levels, respectively, which indicate that the scale development of AMS supply organizations can effectively promote the increase in AGTFP in this province. However, the indirect effect of LSO is significantly negative at the 1% statistical level, indicating that the scaling up of AMS supply organizations in neighboring provinces will inhibit the increase in AGTFP in this province. In contrast, the indirect effect of SPO is statistically significant and positive at the 5% level, indicating that the increased specialization of AMS supply organizations in neighboring provinces will promote the increase in AGTFP in this province. The indirect effects are closely related to the differences in the production and operation modes between LSOs and SPOs. The cross-regional operation of AMS supply organizations is a supplement to the supply of AMS in neighboring provinces based on meeting the demand for AMS in the local province. It forms a complementary relationship with the AMS supply bodies in neighboring provinces. However, with the further expansion of the scale of AMS supply organizations, LSOs mainly involving agricultural enterprises and leading enterprises are formed, equipped with higher-value agricultural machinery and more advanced production technologies. Cross-regional operations of LSOs can achieve a monopoly of the agricultural machinery market in neighboring provinces with cost and price advantages [81], thus causing a severe impact on the agricultural mechanization service system in foreign provinces and ultimately reducing the AGTFP. At the same time, the cross-regional operation of LSOs will also form an AMS system with large-scale households as the core through the coercive effect of service economies of scale. It will accelerate the marketization process of land leasing and force smallholders out of agricultural production [49,82].

		W_1			W_2			W_3	
	DE	IE	TE	DE	IE	TE	DE	IE	TE
LSO	0.192 **	-0.412 ***	-0.219	0.232 **	-0.825 ***	-0.593 **	0.270 ***	-1.270 ***	-1.000 **
	(0.094)	(0.153)	(0.150)	(0.096)	(0.313)	(0.282)	(0.070)	(0.439)	(0.445)
SPO	0.211 ***	0.278 **	0.489 ***	0.209 ***	0.355	0.564 **	0.172 ***	0.442 *	0.614
	(0.068)	(0.123)	(0.157)	(0.067)	(0.233)	(0.259)	(0.065)	(0.239)	(0.254)

 Table 5. Spatial effect decomposition: the organizational level.

Note: Standard errors in parentheses, *, ** and *** indicate significant at the 10%, 5% and 1% statistical levels, respectively.

4.4. The Impact of AMS Supply Individuals on AGTFP

As shown in Table 6, the regression coefficients of LSI ranged from 0.224 to 0.386 under the three spatial weight matrices, and all were statistically significant at the 10% level. This indicates that the scaling of AMS individuals helps promote agricultural green transformation. Meanwhile, the regression coefficient of LSI is higher than that of LSO. This is because the unique advantage of LSI over LSO is that it is nested in the social network of rural society as a member of Chinese rural society. Therefore, LSI can effectively improve the pro-environmental behavior of farmers through the moderating effect of social trust, which in turn can increase AGTFP more efficiently [83]. The regression coefficients of SPI range from -0.398 to -0.311, all were significant at the 1% statistical level. This suggests that the degree of specialization of AMS individuals will inhibit the green transformation of agriculture. There are three main reasons for this. First, the formation of economies of scale in service is the essential condition for AMS to promote the green transformation does not mean a simultaneous scale increase. As shown by statistical data, the number of SPI in China in 2020 is 5.96 times that of LSI, and the value is as high as 330.036 in Guizhou

province. Second, specialized farmers take the provision of AMS as their main source of income and thus may cause excessive input of agricultural machinery elements driven by profit maximization, increasing the undesired output of agricultural production. Third, the value of agricultural machinery owned by specialized farm households is lower than that of LSI. Due to their endowment constraints, they cannot afford to carry out frequent renewal and maintenance of agricultural machinery. Thus, their supply quality of AMS is lower, which in turn causes a decrease in the efficiency of agricultural production factor allocation. In terms of spatial lag variables, the regression coefficients of LSI*W and SPI*W are not significant, which indicates that there is no spatial spillover effect of the effects of LSI and SPI on AGTFP.

	ĩ	<i>N</i> ₁	V	V ₂	V	V ₃
ICI	0.224 *		0.386 **		0.237 *	
L51	(0.130)		(0.164)		(0.138)	
I CI*M	-0.273		-0.523		0.110	
L31 VV	(0.252)		(0.687)		(0.156)	
CDI		-0.398 ***		-0.311 ***		-0.341 ***
511		(0.102)		(0.110)		(0.101)
SDI*W		0.250		1.290		-0.093
511 VV		(0.274)		(0.854)		(0.405)
rho	0.295 ***	0.258 ***	1.813 ***	1.826 ***	0.841 ***	0.786 ***
1110	(0.091)	(0.092)	(0.284)	(0.291)	(0.205)	(0.215)
Control	VES	VES	VES	VES	VES	VES
Variables	TL0	1110	TLO	1L0	110	TL0
Province FE	YES	YES	YES	YES	YES	YES
Time FE	YES	YES	YES	YES	YES	YES
R-squared	0.185	0.293	0.512	0.508	0.451	0.467
Obs	300	300	300	300	300	300

Table 6. The impact of AMS on AGTFP in China: the individual level.

Note: Standard errors in parentheses, *, ** and *** indicate significant at the 10%, 5% and 1% statistical levels, respectively.

This paper further decomposes the spatial effects of LSI and SPI, and the results are shown in Table 7. Under the three spatial weight matrices, the direct effect of LSI is significantly positive at the 10% statistical level, and the direct effect of SPI is significantly negative at the 1% statistical level. This indicates that the scale development of individual AMS significantly increases the AGTFP of the local province, while specialization development suppresses the AGTFP of the local province. Meanwhile, the indirect effects of LSI and SPI are insignificant, which means the LSI and SPI of neighboring provinces do not affect the AGTFP of the local province. The reason for this is that, compared to AMS supply organizations, AMS supply individuals face the dilemma of higher acquisition and maintenance costs of agricultural machinery, insufficient demand matching ability due to information asymmetry, and difficulties in transporting agricultural machinery across regions [65]. Therefore, it is difficult for AMS supply individuals to influence the green transformation of agriculture in other provinces through cross-regional operations.

Table 7. Spatial effect decomposition: the individual level.

		W_1			W_2			W_3	
	DE	IE	TE	DE	IE	TE	DE	IE	TE
LSI	0.238 *	-0.270	-0.032	0.414 **	-0.485	-0.071	0.241 *	0.035	0.277 *
	(0.134)	(0.233)	(0.250)	(0.171)	(0.416)	(0.429)	(0.143)	(0.117)	(0.152)
SPO	-0.406 ***	0.290	-0.116	-0.360 ***	0.906	0.546	-0.336 ***	0.041	-0.296
	(0.105)	(0.251)	(0.265)	(0.109)	(0.565)	(0.485)	(0.104)	(0.305)	(0.330)

Note: Standard errors in parentheses, *, ** and *** indicate significant at the 10%, 5% and 1% statistical levels, respectively.

4.5. Regional Heterogeneity Analysis

Table 8 demonstrates the empirical results of the two-regime spatial Durbin model based on the spatial weight matrix of economic distance. The comparison with the results in Tables 4 and 6 reveals that the significance and the direction of action of the regression coefficients of each core explanatory variable remain consistent, which further strengthens the reliability and stability of the findings of this study. On this basis, we focus on the asymmetry of spatial spillover effects among different zone systems. From the results of d¹, the regression coefficient of rho1 is negative but not significant, the regression coefficient of rho2 is significantly positive at the 1% statistical level, and the regression coefficient of rho1-rho2 is significantly negative at the 5% statistical level. This indicates an asymmetry in the spatial spillover effect of AGTFP, and the spatial spillover effect is significantly higher in non-main grain production areas than in main grain production areas. There are two main reasons for this. First, the crop sowing area in the main grain production areas is vast, and the terrain is flat, so the demand for AMS is more significant and more favorable for agricultural machinery operation. With the rapid development of AMS, the main grain production areas with massive market scale can easily form service scale economy and operation scale economy through the horizontal and vertical division of labor, significantly improving AMS supply agents' profit. Therefore, the AMS supply agents in the main grain production areas are more inclined to prioritize meeting the demand for AMS in the local province. Second, from 2011 to 2020, the average value of SPO in non-main grain production areas increased from 0.2430 to 0.5370, and the average value of LPO increased from 0.1514 to 0.3941, while the average value of SPO in grainproducing regions increased from 0.2634 to 0.4536, and the average value of LPO increased from 0.1967 to 0.3542. Therefore, compared with non-main grain production areas, the average value of LPO increased from 0.1514 to 0.3941. Although they have more total AMS supply organizations, main grain production areas do not show significant advantages in developing AMS specialization and large-scale and even gradually lag from the early lead.

Table 8. Empirical results of heterogeneity analysis.

			d^1			à	2	
LSO	0.171 *** (2.612)				0.173 *** (2.587)			
LSO*W	-0.180 ** (-2.140)				-0.183 ** (-2.201)			
SPO		0.135 * (1.886)				0.131 * (1.746)		
SPO*W		0.308 * (1.866)				0.286 * (1.730)		
LSI			0.546 *** (3.281)				0.597 *** (3.526)	
LSI*W			-0.035 (-0.033)				0.318 (0.305)	
SPI				-0.234 *** (-2.728)				-0.223 *** (-2.584)
SPI*W				0.441 (1.159)				0.439 (1.132)
rho1	-0.157 (-1.250)	-0.177 (-1.405)	-0.103 (-0.813)	-0.167 (-1.298)	0.272 *** (2.982)	0.269 *** (2.972)	0.272 *** (2.947)	0.272 *** (2.947)
rho2	0.410 *** (3.303)	0.405 *** (3.268)	0.348 *** (2.776)	0.411 *** (3.289)	-0.197 (-0.986)	-0.223 (1.110)	-0.218 (-1.078)	-0.218 (-1.078)
rho1-rho2	-0.567 *** (-3.173)	-0.582 *** (-3.259)	-0.452 ** (-2.504)	-0.578 *** (-3.169)	0.469 ** (2.129)	0.493 ** (2.227)	0.491 ** (2.194)	0.491 ** (2.194)
Control Variables	YES	YES	YES	YES	YES	YES	YES	YES

0.847

300

Time FE

R-squared Obs

		Table 8. Co	nt.						
d^1						d	2		-
Province FE	YES	YES	YES	YES	YES	YES	YES	YES	
Time FE	YES	YES	YES	YES	YES	YES	YES	YES	

0.853

300

0.845

300

0.843

300

300 Note: T-value in parentheses, *, ** and *** indicate significant at the 10%, 5% and 1% statistical levels, respectively.

0.847

0.844

300

0.847

300

From the results of d^2 , the regression coefficients of rho1 are all significantly positive at the 1% statistical level, the regression coefficients of rho2 are not significant, and the regression coefficients of rho1-rho2 are all significantly positive at the 5% statistical level. This indicates that the spatial spillover effect of AGTFP is asymmetric among provinces with different levels of economic development, and the spatial spillover effect is significantly higher in economically developed provinces than in other provinces. This is because economically developed provinces have the advantage of transportation infrastructure, which helps to promote the free flow of factors between regions [84,85]. Therefore, it helps to reduce the mechanical transportation costs of AMS supply organizations when operating across regions, thus forming an asymmetry of the spatial spillover effect. In addition to this, compared with other provinces, economically developed provinces have a higher level of scientific research investment and promotion of agricultural machinery [86], which provides essential support for AMS to exert spatial spillover effects.

5. Discussion

For countries constrained by the pattern of "smallholder management" due to the fragmentation of land and unclear land property rights, attempts to rely on economies of scale to deepen the horizontal division of labor in agriculture and thus promote the agricultural green transformation are often ineffective. In contrast, the construction of AMS system can deepen the vertical division of labor in agriculture through the realization of the service scale economy and thus become another effective path to agricultural green transformation in economies with smallholders as the mainstay of agricultural production. This paper's theoretical contributions and innovations are mainly reflected in the following three aspects by comparing the existing studies.

5.1. Development Path of AMS under the Goal of Agricultural Green Transformation

The existing studies have demonstrated the critical role of AMS in improving agricultural productivity and environmental efficiency at both macro and micro levels [42,44,54,55,57,83]. These findings provide an essential theoretical basis for AMS's positive role in agriculture's green transformation. However, the issue of how to promote the rapid development of AMS remains controversial. The reason for this is that most studies have measured the development of AMS in terms of AMS production value and farmers' decision to purchase AMS. The resulting policy insights are focused on two general aspects of promoting the rapid development of AMS and improving farmers' responsiveness, which is challenging to form practical guidance for the rapid development of AMS to achieve the goal of agricultural green transformation. Unlike the existing studies, this paper constructs indicators related to the development of AMS in terms of specialization and the degree of scale from the multiple service supply agents in the Chinese AMS market. The results show that the impact of different AMS supply agents on AGTFP shows apparent differentiation. Among them, the impact of SPO, LSO, and LSI on AGTFP are significantly positive, but the effects of SPI on AGTFP are significantly negative. This key finding provides essential insights into the development path of China's AMS system under the goal of green transformation in agriculture, namely, to increase the support of AMS supply organizations and actively promote the transformation of AMS supply individuals.

0.853

300

Expressly, compared to AMS supply organizations, AMS supply individuals are limited by their inadequate endowments, which make it difficult for them to acquire high-value agricultural machinery and have relatively low management skills due to the lack of training [87]. Meanwhile, although China has been implementing the agricultural machinery acquisition subsidy policy for a long time, the beneficiaries of this policy are mainly large-scale operators, not small-scale operators [88]. Therefore, due to endowment constraints and policy exclusion, it is difficult for AMS supply individuals to provide high-quality AMS with green agricultural attributes. Based on this, actively promoting the transformation of AMS supply individuals becomes the primary issue in constructing the development path of AMS. This paper presents two transformation possibilities combined with China's agricultural development practice. First, with the rapid rise of labor prices, the wage income that can be brought by non-farm employment exceeds the operating income by providing AMS. The rapid labor market development provides an opportunity for the transformation of AMS supply individuals. Therefore, it is possible to promote the withdrawal of some agricultural machinery professionals from the AMS market by increasing the non-agricultural transfer of rural labor. This initiative can eliminate some old and low-value agricultural machinery to improve the quality of AMS supply and give up part of the AMS market demand to cultivate other AMS supply agents. Secondly, farmers' cooperatives have played an essential role in China's rural economic development by building a close interest linkage mechanism among smallholders through the "weakweak association" [89,90]. Similarly, agricultural machinery professional cooperatives are an essential part of the AMS supply organization and have played an important role in promoting the agricultural green transformation. Therefore, AMS supply individuals with high-value agricultural machinery can be actively fostered to transform into AMS supply organizations in a "weak-weak association" approach.

5.2. Characteristics of AMS Cross-Regional Operations

There has been a debate on the relationship between agricultural mechanization and the green transformation of agriculture. Specifically, some scholars believe that the negative environmental impact of energy consumption caused by agricultural mechanization cannot be ignored. In contrast, some scholars believe that the carbon reduction effect of agricultural mechanization will offset or even reverse its negative environmental impact. The main reason for this debate is that some existing studies ignore the critical impact mechanism of cross-regional operation of agricultural mechanization [42,80]. Meanwhile, the impact of AMS on AGTFP will be underestimated because AMS operating across regions is not counted in the total AMS of neighboring provinces. With the promotion of agricultural mechanization, the scale of cross-regional operation of AMS in China has become larger and larger. In 2020, the area of cross-regional operation of agricultural machinery in China was 19899.67 khm², accounting for 35.22% of the area operated by agricultural machinery cooperatives. Therefore, the cross-regional operation has become an essential channel for AMS to promote agricultural green transformation. Based on this, this paper uses a spatial Durbin model test to find a significant spatial spillover effect of AMS on AGTFP. This finding further demonstrates the critical reality that AMS promotes agricultural green transformation in neighboring provinces through cross-regional operations.

The analysis of the spatial lag term found that the spatial spillover effect of AMS supply organizations was significant. However, the spatial spillover effect of AMS supply individuals was not significant. This finding suggests that the cross-regional operation of AMS is mainly driven by AMS supply organizations, similar to the findings of Fang et al. [66] and Wu et al. [67]. Unlike existing studies, this paper further finds that the spatial spillover effect of specialization of AMS supply organizations on AGTFP is significantly positive. In contrast, the spillover effect of large-scale AMS supply organizations is significantly negative. This finding suggests that a single-minded push for rapid AMS development does not effectively promote a green transformation of agriculture. Local government should regulate the cross-regional operation behavior of large-scale AMS supply organizations

is necessary. Furthermore, it is also necessary to cultivate AMS supply agents in the local province through agricultural machinery purchase subsidy policies. These initiatives can counteract the impact of large-scale AMS supply organizations from other regions on the agricultural mechanization service market in the local province.

6. Conclusions

This study examines the development of AMS in China and explores how economies with smallholder as the mainstay of agricultural production can construct a transition path to green agriculture. Based on the panel data of 30 provinces in China from 2011 to 2020, this paper measures the AGTFP using the Super-SBM model. Based on this, using the spatial Durbin model and the two-regime spatial Durbin model, this paper tests the influence and spatial effect of different AMS supply agents on AGTFP. The main conclusions are as follows.

First, during the study period, China's AGTFP showed a stable growth trend, with the mean value increasing from 0.1990 in 2011 to 0.5590 in 2020. Among them, the AGTFP of Beijing and Hainan is relatively higher. In contrast, the AGTFP of Jilin and Inner Mongolia are always lower. By stage, compared with the 12th Five-Year Plan period (2011–2015), the average value and the growth rate of AGTFP in the 13th Five-Year Plan period (2016–2020) is higher and faster. By region, agricultural green transformation is faster in non-main grain production and economically developed regions. In contrast, agricultural green transformation tasks are relatively heavier in food-producing and less economically developed regions.

Second, AMS is an essential factor influencing the growth of AGTFP, but there are still differences among different supplying agents. At the organizational level, the degree of specialization and large-scale of AMS supply organizations will significantly contribute to the increase in AGTFP of the local province. However, the cross-regional operation of specialized AMS supply organizations will significantly increase the AGTFP of neighboring provinces. In contrast, the cross-regional operation of large-scale AMS supply organizations will significantly decrease the AGTFP of neighboring provinces. At the individual level, the degree of specialization of AMS supply individual will significantly decrease the AGTFP of the local province. In contrast, the large-scale MAS supply individual has the opposite effect. At the same time, individual AMS supply agents cannot operate across regions.

Third, the spatial spillover effect of AGTFP is asymmetric due to the significant differences in the supply of AMS among different regions. Specifically, the spatial spillover effect of AGTFP is significantly higher in non-main grain production areas than in main grain production areas. In comparison, the spatial spillover effect of AGTFP is significantly higher in economically developed areas than in less developed areas. Meanwhile, the spatial spillover effect asymmetry also reflects the flow direction of AMS resources.

Based on the above conclusions, to further promote the agricultural green transformation in China, this paper provides the following policy implications. First, promoting the popularization of agricultural mechanization throughout agricultural production. It will help to provide an essential foundation for the rapid development of the AMS market. Meanwhile, with the goal of green transformation of agriculture, it is necessary to enhance agricultural machinery's R&D and production capacity and promote clean energy-based agricultural machinery to replace traditional fossil energy-based agricultural machinery. Second, actively promote the transformation of AMS supply individuals to AMS supply organizations. Encourage small farmers to join together extensively to realize the transformation through the "farmers' cooperative." It can improve the organization of AMS supply subjects. Third, improve the service quality of AMS supply organizations. Increase the agricultural machinery purchase subsidies for AMS supply organizations to improve the service quality and agricultural machinery scale of AMS supply organizations. Meanwhile, strengthen the construction of agricultural mechanization personnel to improve the management capacity of AMS supply organizations. Fourth, the main grain-producing and less economically developed regions should accelerate the construction of local AMS systems. The central government needs to provide corresponding financial support, resource inclination, and legal regulation to guarantee the construction of the AMS systems in critical regions. It will help to avoid the unbalanced and insufficient process of agricultural green transformation among regions.

The main limitations of this paper are the following three aspects. First, based on provincial panel data, this study finds a spatial spillover effect of the cross-regional operation behavior of AMS supply organizations on the impact of AGTFP. However, although AMS supply individuals cannot operate across regions between provinces, it is still being determined whether they operate across regions between smaller administrative units such as counties and communes. Therefore, future research can explore more deeply the crossregional operation characteristics of different AMS supply individuals based on county panel data. Second, combining statistical data and existing studies [56,65], AMS in China has developed rapidly since 2004. Unfortunately, limited by data availability issues, this study only examined the relationship between AMS and AGTFP during 2010–2020 and has yet to provide a clear picture of the full development of AMS in China. Therefore, longer-term panel data could be considered in future studies to more accurately capture the relationship between the development of AMS and the agricultural green transformation. Third, AMS exists in many aspects of agricultural production, such as tillage, sowing, irrigation, and harvesting [43,64]. A possible future research direction is to explore in depth the heterogeneous role of AMS in different product segments in promoting the agricultural green transformation, which will help to construct a more comprehensive AMS system.

Author Contributions: Conceptualization, X.L. and R.G.; methodology, R.G.; validation, R.G.; formal analysis, X.L.; data curation, X.L.; writing—original draft preparation, X.L.; writing—review and editing, R.G.; visualization, X.L.; supervision, R.G.; project administration, X.L. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by Anhui Provincial Philosophy and Social Science Planning Project, grant number AHSKQ2020D17; Key projects of humanities and social sciences of Anhui Provincial Department of Education, grant number 2022AH051615 and 2022AH040230.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The data presented in this study are available upon request from the corresponding author.

Acknowledgments: The authors would like to thank the editor and reviewers for their insightful comments and suggestions.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Shen, Z.; Baležentis, T.; Ferrier, G.D. Agricultural productivity evolution in China: A generalized decomposition of the Luenberger-Hicks-Moorsteen productivity indicator. *China Econ. Rev.* 2019, *57*, 101315. [CrossRef]
- Zhang, H.; Zhang, H.; Li, W.; Li, G. Agricultural Modernization with Chinese Characteristics: Target Orientation, Reform and Innovation. *Chin. Rural Econ.* 2015, 1, 4–13.
- 3. Zou, L.; Wang, Y.; Liu, Y. Spatial-temporal evolution of agricultural ecological risks in China in recent 40 years. *Environ. Sci. Pollut. Res. Int.* **2022**, *29*, 3686–3701. [CrossRef] [PubMed]
- 4. Gao, X.; Li, B.; Jiang, S.; Nie, Y. Can Increasing Scale Efficiency Curb Agricultural Nonpoint Source Pollution? *Int. J. Environ. Res. Public Health* **2021**, *18*, 8798. [CrossRef]
- Han, H.; Zhong, Z.; Guo, Y.; Xi, F.; Liu, S. Coupling and decoupling effects of agricultural carbon emissions in China and their driving factors. *Environ. Sci. Pollut. Res. Int.* 2018, 25, 25280–25293. [CrossRef]
- Wen, S.; Hu, Y.; Liu, H. Measurement and Spatial–Temporal Characteristics of Agricultural Carbon Emission in China: An Internal Structural Perspective. *Agriculture* 2022, 12, 1749. [CrossRef]
- Liu, Y.; Zhou, Y. Reflections on China's food security and land use policy under rapid urbanization. Land Use Policy 2021, 109, 105699. [CrossRef]
- 8. Tilman, D.; Balzer, C.; Hill, J.; Befort, B.L. Global food demand and the sustainable intensification of agriculture. *Proc. Natl. Acad. Sci. USA* **2011**, *108*, 20260–20264. [CrossRef]

- Shiferaw, B.; Smale, M.; Braun, H.-J.; Duveiller, E.; Reynolds, M.; Muricho, G. Crops that feed the world 10. Past successes and future challenges to the role played by wheat in global food security. *Food Secur.* 2013, *5*, 291–317. [CrossRef]
- 10. Pingali, P.L. Green revolution: Impacts, limits, and the path ahead. Proc. Natl. Acad. Sci. USA 2012, 109, 12302–12308. [CrossRef]
- 11. Zhang, T. Why are Chinese small farmers planting more grain crops?—Review of the Logic of Grain Planting (2018). *China Agric. Econ. Rev.* **2019**, *12*, 173–175. [CrossRef]
- 12. Ji, X.; Rozelle, S.; Huang, J.; Zhang, L.; Zhang, T. Are China's Farms Growing? China World Econ. 2016, 24, 41–62. [CrossRef]
- 13. Cui, Z.; Zhang, H.; Chen, X.; Zhang, C.; Ma, W.; Huang, C.; Zhang, W.; Mi, G.; Miao, Y.; Li, X.; et al. Pursuing sustainable productivity with millions of smallholder farmers. *Nature* **2018**, *555*, 363–366. [CrossRef]
- 14. Gao, J.; Song, G.; Sun, X. Does labor migration affect rural land transfer? Evidence from China. *Land Use Policy* **2020**, *99*, 105096. [CrossRef]
- 15. Sims, B.; Kienzle, J. Sustainable Agricultural Mechanization for Smallholders: What Is It and How Can We Implement It? *Agriculture* **2017**, *7*, 50. [CrossRef]
- Sims, B.; Heney, J. Promoting Smallholder Adoption of Conservation Agriculture through Mechanization Services. *Agriculture* 2017, 7, 64. [CrossRef]
- 17. Luo, B. Service Scale Management: Vertical Division of Labor, Horizontal Division of Labor and Specialization of Connected Farmland. *Chin. Rural Econ.* **2017**, *11*, 2–16.
- Yu, D.; Liu, L.; Gao, S.; Yuan, S.; Shen, Q.; Chen, H. Impact of carbon trading on agricultural green total factor productivity in China. J. Clean. Prod. 2022, 367, 132789. [CrossRef]
- Fang, L.; Hu, R.; Mao, H.; Chen, S. How crop insurance influences agricultural green total factor productivity: Evidence from Chinese farmers. J. Clean. Prod. 2021, 321, 128977. [CrossRef]
- 20. Huang, X.; Feng, C.; Qin, J.; Wang, X.; Zhang, T. Measuring China's agricultural green total factor productivity and its drivers during 1998-2019. *Sci. Total Environ.* 2022, 829, 154477. [CrossRef]
- 21. Liu, S.; Lei, P.; Li, X.; Li, Y. A nonseparable undesirable output modified three-stage data envelopment analysis application for evaluation of agricultural green total factor productivity in China. *Sci. Total Environ.* **2022**, *838*, 155947. [CrossRef] [PubMed]
- 22. Ge, P.; Wang, S.; Huang, X. Measurement for China's Agricultural Green TFP. China Popul. Resour. Environ. 2018, 28, 66–74.
- 23. Xiao, S.; He, Z.; Zhang, W.; Qin, X. The Agricultural Green Production following the Technological Progress: Evidence from China. *Int. J. Environ. Res. Public Health* **2022**, *19*, 9876. [CrossRef]
- 24. Diao, P.; Zhang, Z.; Jin, Z. Dynamic and static analysis of agricultural productivity in China. *China Agric. Econ. Rev.* **2018**, 10, 293–312. [CrossRef]
- 25. Liu, D.; Zhu, X.; Wang, Y. China's agricultural green total factor productivity based on carbon emission: An analysis of evolution trend and influencing factors. *J. Clean. Prod.* 2021, 278, 123692. [CrossRef]
- Xu, X.; Huang, X.; Huang, J.; Gao, X.; Chen, L. Spatial-Temporal Characteristics of Agriculture Green Total Factor Productivity in China, 1998-2016: Based on More Sophisticated Calculations of Carbon Emissions. *Int. J. Environ. Res. Public Health* 2019, 16, 3932. [CrossRef]
- Zhu, L.; Shi, R.; Mi, L.; Liu, P.; Wang, G. Spatial Distribution and Convergence of Agricultural Green Total Factor Productivity in China. Int. J. Environ. Res. Public Health 2022, 19, 8786. [CrossRef]
- Liu, Y.; Feng, C. What drives the fluctuations of "green" productivity in China's agricultural sector? A weighted Russell directional distance approach. *Resour. Conserv. Recycl.* 2019, 147, 201–213. [CrossRef]
- Zhong, S.; Li, Y.; Li, J.; Yang, H. Measurement of total factor productivity of green agriculture in China: Analysis of the regional differences based on China. *PLoS ONE* 2021, *16*, e0257239. [CrossRef]
- Ablimit, R.; Li, W.; Zhang, J.; Gao, H.; Zhao, Y.; Cheng, M.; Meng, X.; An, L.; Chen, Y. Altering microbial community for improving soil properties and agricultural sustainability during a 10-year maize-green manure intercropping in Northwest China. *J. Environ. Manag.* 2022, 321, 115859. [CrossRef]
- Zhu, Y.; Zhang, Y.; Piao, H. Does Agricultural Mechanization Improve the Green Total Factor Productivity of China's Planting Industry? *Energies* 2022, 15, 940. [CrossRef]
- 32. Liu, F.; Lv, N. The threshold effect test of human capital on the growth of agricultural green total factor productivity: Evidence from China. *Int. J. Electr. Eng. Educ.* **2021**, *in press*. [CrossRef]
- Ren, C.; Liu, S.; van Grinsven, H.; Reis, S.; Jin, S.; Liu, H.; Gu, B. The impact of farm size on agricultural sustainability. J. Clean. Prod. 2019, 220, 357–367. [CrossRef]
- 34. Sun, Y. Environmental regulation, agricultural green technology innovation, and agricultural green total factor productivity. *Front. Environ. Sci.* **2022**, *10*, 955954. [CrossRef]
- 35. Gao, Q.; Cheng, C.; Sun, G.; Li, J. The Impact of Digital Inclusive Finance on Agricultural Green Total Factor Productivity: Evidence From China. *Front. Ecol.* 2022, *10*, 905644. [CrossRef]
- 36. Deng, H.; Jing, X.; Shen, Z. Internet technology and green productivity in agriculture. *Environ. Sci. Pollut. Res. Int.* 2022, 29, 81441–81451. [CrossRef]
- 37. Li, H.; Zhou, X.; Tang, M.; Guo, L. Impact of Population Aging and Renewable Energy Consumption on Agricultural Green Total Factor Productivity in Rural China: Evidence from Panel VAR Approach. *Agriculture* **2022**, *12*, 715. [CrossRef]
- 38. Yang, Y.; Ma, H.; Wu, G. Agricultural Green Total Factor Productivity under the Distortion of the Factor Market in China. *Sustainability* **2022**, *14*, 9309. [CrossRef]

- 39. Xu, P.; Jin, Z.; Ye, X.; Wang, C. Efficiency Measurement and Spatial Spillover Effect of Green Agricultural Development in China. *Front. Environ. Sci.* **2022**, *10*, 909321. [CrossRef]
- 40. Song, Y.; Zhang, B.; Wang, J.; Kwek, K. The impact of climate change on China's agricultural green total factor productivity. *Technol. Forecast. Soc. Change* **2022**, *185*, 122054. [CrossRef]
- Hu, J.; Zhang, X.; Wang, T. Spatial Spillover Effects of Resource Misallocation on the Green Total Factor Productivity in Chinese Agriculture. Int. J. Environ. Res. Public Health 2022, 19, 15718. [CrossRef] [PubMed]
- 42. Xu, Q.; Zhu, P.; Tang, L. Agricultural Services: Another Way of Farmland Utilization and Its Effect on Agricultural Green Total Factor Productivity in China. *Land* **2022**, *11*, 1170. [CrossRef]
- 43. Zhu, Y.; Deng, J.; Wang, M.; Tan, Y.; Yao, W.; Zhang, Y. Can Agricultural Productive Services Promote Agricultural Environmental Efficiency in China? *Int. J. Environ. Res. Public Health* **2022**, *19*, 9339. [CrossRef] [PubMed]
- Li, C.; Xu, J.; Wang, Y. Can Socialized Service of Agricultural Green Production Improve Agricultural Green Productivity? J. Agrotech. Econ. 2021, 9, 36–49. [CrossRef]
- Zhang, M.; Tong, T.; Chen, Z. Can Agricultural Socialized Service Improve Agricultural Green Productivity? *South China J. Econ.* 2022, *in press*. [CrossRef]
- 46. Otsuka, K. Food insecurity, income inequality, and the changing comparative advantage in world agriculture. *Agric. Econ.* **2013**, 44, 7–18. [CrossRef]
- Zhang, X.; Yang, J.; Thomas, R. Mechanization outsourcing clusters and division of labor in Chinese agriculture. *China Econ. Rev.* 2017, 43, 184–195. [CrossRef]
- Qian, L.; Lu, H.; Gao, Q.; Lu, H. Household-owned farm machinery vs. outsourced machinery services: The impact of agricultural mechanization on the land leasing behavior of relatively large-scale farmers in China. *Land Use Policy* 2022, *115*, 106008. [CrossRef]
- 49. Zhou, J. The Dual Division of Agricultural Socialization Service and its Consequences under the Background of Land Circulation. J. Nanjing Agric. Univ. 2017, 17, 141–151+166.
- 50. Anang, B.T.; Asante, B.O. Farm household access to agricultural services in northern Ghana. Heliyon 2020, 6, e05517. [CrossRef]
- 51. Lu, Q.; Du, X.; Qiu, H. Adoption patterns and productivity impacts of agricultural mechanization services. *Agric. Econ.* **2022**, *53*, 826–845. [CrossRef]
- 52. Guo, L.; Duan, X.; Li, H.; Yang, W.; Ren, Y.; Guo, Y. Does a higher minimum wage accelerate labour division in agricultural production? Evidence from the main rice-planting area in China. *Econ. Res.* **2021**, *35*, 2984–3010. [CrossRef]
- Liao, W.; Zeng, F.; Chanieabate, M. Mechanization of Small-Scale Agriculture in China: Lessons for Enhancing Smallholder Access to Agricultural Machinery. Sustainability 2022, 14, 7964. [CrossRef]
- 54. Deng, X.; Xu, D.; Zeng, M.; Qi, Y. Does outsourcing affect agricultural productivity of farmer households? Evidence from China. *China Agric. Econ. Rev.* **2020**, *12*, 673–688. [CrossRef]
- 55. Tang, L.; Liu, Q.; Yang, W.; Wang, J. Do agricultural services contribute to cost saving? Evidence from Chinese rice farmers. *China Agric. Econ. Rev.* **2018**, *10*, 323–337. [CrossRef]
- Qiu, T.; Luo, B. Do small farms prefer agricultural mechanization services? Evidence from wheat production in China. *Appl. Econ.* 2021, 53, 2962–2973. [CrossRef]
- 57. Mi, Q.; Li, X.; Gao, J. How to improve the welfare of smallholders through agricultural production outsourcing: Evidence from cotton farmers in Xinjiang, Northwest China. J. Clean. Prod. 2020, 256, 120636. [CrossRef]
- 58. Lyne, M.C.; Jonas, N.; Ortmann, G.F. A quantitative assessment of an outsourced agricultural extension service in the Umzimkhulu District of KwaZulu-Natal, South Africa. *J. Agric. Educ. Ext.* **2017**, *24*, 51–64. [CrossRef]
- 59. Houssou, N.; Diao, X.; Cossar, F.; Kolavalli, S.; Jimah, K.; Aboagye, P.O. Agricultural Mechanization in Ghana: Is Specialized Agricultural Mechanization Service Provision a Viable Business Model? *Am. J. Agric. Econ.* **2013**, *95*, 1237–1244. [CrossRef]
- 60. Qiu, T.; Choy, S.; Li, Y.; Luo, B.; Li, J. Farmers' Exit from Land Operation in Rural China: Does the Price of Agricultural Mechanization Services Matter? *China World Econ.* **2021**, *29*, 99–122. [CrossRef]
- 61. Chi, L.; Han, S.; Huan, M.; Li, Y.; Liu, J. Land Fragmentation, Technology Adoption and Chemical Fertilizer Application: Evidence from China. *Int. J. Environ. Res. Public Health* **2022**, *19*, 8147. [CrossRef] [PubMed]
- 62. Li, Q.; Wang, J.; Wu, J.; Zhai, Q. The dual impacts of specialized agricultural services on pesticide application intensity: Evidence from China. *Pest. Manag. Sci.* 2022, *79*, 76–87. [CrossRef] [PubMed]
- Aguilera, E.; Guzmán, G.I.; González de Molina, M.; Soto, D.; Infante-Amate, J. From animals to machines. The impact of mechanization on the carbon footprint of traction in Spanish agriculture: 1900–2014. J. Clean. Prod. 2019, 221, 295–305. [CrossRef]
- 64. Qing, Y.; Chen, M.; Sheng, Y.; Huang, J. Mechanization services, farm productivity and institutional innovation in China. *China Agric. Econ. Rev.* **2019**, *11*, 536–554. [CrossRef]
- 65. Yang, J.; Huang, Z.; Zhang, X.; Reardon, T. The Rapid Rise of Cross-Regional Agricultural Mechanization Services in China. *Am. J. Agric. Econ.* **2013**, *95*, 1245–1251. [CrossRef]
- Fang, S.; Wei, L.; Wu, J. The Spatial Spillover Effect of Agricultural Mechanization and Its Distribution Pattern: The Perspective of Interregional-Service of Agricultural Machinery. J. Manag. World 2017, 11, 65–78+187–188. [CrossRef]
- 67. Wu, J.; Fang, S.; LI, G.; Xu, G. The Spillover Effect of Agricultural Mechanization on Grain Output in China: From the Perspective of Cross-regional Mechanization Service. *Chin. Rural Econ.* **2017**, *6*, 44–57.
- Li, J.; Lin, Q. Can the Adjustment of China's Grain Purchase and Storage Policy Improve Its Green Productivity? Int. J. Environ. Res. Public Health 2022, 19, 6310. [CrossRef]

- 69. Wang, L.; Tang, J.; Tang, M.; Su, M.; Guo, L. Scale of Operation, Financial Support, and Agricultural Green Total Factor Productivity: Evidence from China. *Int. J. Environ. Res. Public Health* **2022**, *19*, 9043. [CrossRef]
- 70. Razzaq, A.; Qing, P.; Naseer, M.; Abid, M.; Anwar, M.; Javed, I. Can the informal groundwater markets improve water use efficiency and equity? Evidence from a semi-arid region of Pakistan. *Sci. Total Environ.* **2019**, *666*, 849–857. [CrossRef]
- Jomthanachai, S.; Wong, W.-P.; Lim, C.-P. An Application of Data Envelopment Analysis and Machine Learning Approach to Risk Management. *IEEE Access* 2021, 9, 85978–85994. [CrossRef]
- 72. Hong, H.K.; Ha, S.H.; Shin, C.K.; Park, S.C.; Kima, S.H. Evaluating the efficiency of system integration projects using data envelopment analysis (DEA) and machine learning. *Expert Syst. Appl.* **1999**, *16*, 283–296. [CrossRef]
- 73. Yang, N.; Sun, X.; Qi, Q. Impact of factor quality improvement on agricultural carbon emissions: Evidence from China's high-standard farmland. *Front. Environ. Sci.* 2022, 10, 989684. [CrossRef]
- Huang, X.; Xu, X.; Wang, Q.; Zhang, L.; Gao, X.; Chen, L. Assessment of Agricultural Carbon Emissions and Their Spatiotemporal Changes in China, 1997–2016. Int. J. Environ. Res. Public Health 2019, 16, 3105. [CrossRef]
- 75. Zhang, L.; Pang, J.; Chen, X.; Lu, Z. Carbon emissions, energy consumption and economic growth: Evidence from the agricultural sector of China's main grain-producing areas. *Sci. Total Environ.* **2019**, *665*, 1017–1025. [CrossRef]
- Yang, H.; Wang, X.; Bin, P. Agriculture carbon-emission reduction and changing factors behind agricultural eco-efficiency growth in China. J. Clean. Prod. 2022, 334, 130193. [CrossRef]
- 77. IPCC. Climate Change 2007: Mitigation: Contribution of Working Group III to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change: Summary for Policymakers and Technical Summary; Cambridge University Press: Cambridge, UK, 2007.
- Elhorst, J.; Freret, S. Evidence of Political Yardstick Competition in France Uing A Two-Regime Spatial Durbin Model with Fixed Effects. J. Reg. Sci. 2009, 49, 931–951. [CrossRef]
- 79. Ma, G.; Lv, D.; Luo, Y.; Jiang, T. Environmental Regulation, Urban-Rural Income Gap and Agricultural Green Total Factor Productivity. *Sustainability* **2022**, *14*, 8995. [CrossRef]
- Chi, Y.; Zhou, W.; Wang, Z.; Hu, Y.; Han, X. The Influence Paths of Agricultural Mechanization on Green Agricultural Development. Sustainability 2022, 13, 12984. [CrossRef]
- Zhou, Y.; Wang, O.; Tang, K. Cost-effective Analysis on Machinery Operation of China Agricultural Machinery Cooperitives: Based on Comparision of East-Central-West Regions and Different Scales. J. Nanjing Agric. Univ. 2015, 15, 51–56+126.
- 82. Qiu, T.; Shi, X.; He, Q.; Luo, B. The paradox of developing agricultural mechanization services in China: Supporting or kicking out smallholder farmers? *China Econ. Rev.* 2021, *69*, 101680. [CrossRef]
- 83. Yang, C.; Zeng, H.; Zhang, Y. Are Socialized Services of Agricultural Green Production Conducive to the Reduction in Fertilizer Input? Empirical Evidence from Rural China. *Int. J. Environ. Res. Public Health* **2022**, *19*, 14856. [CrossRef] [PubMed]
- 84. Hui, E.C.M.; Li, X.; Chen, T.; Lang, W. Deciphering the spatial structure of China's megacity region: A new bay area—The Guangdong-Hong Kong-Macao Greater Bay Area in the making. *Cities* **2020**, *105*, 102168. [CrossRef]
- Zhong, C.; Hu, R.; Wang, M.; Xue, W.; He, L. The impact of urbanization on urban agriculture: Evidence from China. J. Clean. Prod. 2020, 276, 122686. [CrossRef]
- Deng, H.; Jin, Y.; Pray, C.; Hu, R.; Xia, E.; Meng, H. Impact of public research and development and extension on agricultural productivity in China from 1990 to 2013. *China Econ. Rev.* 2021, 70, 101699. [CrossRef]
- 87. Huo, Y.; Ye, S.; Wu, Z.; Zhang, F.; Mi, G. Barriers to the Development of Agricultural Mechanization in the North and Northeast China Plains: A Farmer Survey. *Agriculture* **2022**, *12*, 287. [CrossRef]
- Huang, J.; Wang, X.; Rozelle, S. The subsidization of farming households in China's agriculture. *Food Policy* 2013, 41, 124–132.
 [CrossRef]
- 89. Huang, Z.; Liang, Q. Agricultural organizations and the role of farmer cooperatives in China since 1978: Past and future. *China Agric. Econ. Rev.* **2018**, *10*, 48–64. [CrossRef]
- Liu, Z.; Yang, D.; Wen, T. Agricultural production mode transformation and production efficiency. *China Agric. Econ. Rev.* 2018, 11, 160–179. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.